Title: Cognitive fatigue in relation to depressive symptoms after treatment for childhood cancer

Authors:

Elin Irestorm (elin.irestorm@med.lu.se)

Ingrid Tonning-Olsson (ingrid.tonning_olsson@psy.lu.se)

Birgitta Johansson (birgitta.johannson@neuro.gu.se)

Ingrid Ora (ingrid.ora@med.lu.se)

Version: 3 Date: 14 Mar 2020

Author’s response to reviews:

Dear Dr Thorsteinsson,

Thank you for your decision, and for getting back to us so quickly with your feedback. Changes have been made in the manuscript, tables and reference list according to your comments.

Manuscript: The typo on page 6 line 35 has been corrected. Reports of SD and CI have been changed according to instructions (and p-values removed).

Tables: The tables have been updated so that headers have been changed and horizontal lines have been added under the captions. For Table 2, the headings “BT” and “ALL” are now spelled out. (This was not requested for the other tables, but due to small cell size in the other tables, Table 2 was the only one where it was possible). SD and M are now in italics and have been removed from the notes. We settled on using "participant characteristics" rather than "patient characteristics" for Table 1, though, as half of the participants were not really patients.

Reference list: Doi has been changed in reference 44. Page numbers have been corrected for reference 13 and 18. All references have been changed so that those journals that have abbreviated names are now abbreviated. Capital letters have been removed after colon in four references.

We are looking forward to hearing from you again.

On behalf of all authors,

Elin Irestorm